\begin{table}%t3
\caption{Interferometric and photometric data used in the present paper.
The references for the interferometric measurements are:
Nordgren et~al. (\cite{nordgren00}, N00),
Mourard et~al. (\cite{mourard97}, M97),
Nordgren et~al. (\cite{nordgren02}, N02),
Lane et~al. (\cite{lane02}, L02), and
Kervella et~al. (\cite{kervella04a}, K04).
JD is the Julian date of the measurement,
$\lambda$ the interferometric measurement wavelength (in~$\mu$m),
$\phi$ the phase, $\theta_{\rm UD}$ the uniform disk and~$\theta_{\rm LD}$ the limb darkened angular diameters (in~mas). The magnitudes are corrected for interstellar extinction
(see Sect.~\ref{sect_extinct}).}
\label{table_measurements1}
%\centering

\begin{tabular}{lcccccccccccc}
\hline \hline 
Star & Ref. & JD & $\lambda$ & $\phi$ & $\theta_{\rm LD}$ & $B_0$ & $V_0$ & $R_0$ & $I_0$ & $J_0$ & $H_0$ & $K_0$ \\
\hline 
                         $\alpha$~UMi & N00 & avg & 0.74 & avg & $ 3.284 \pm 0.021 $ & 2.66 & 1.99 & 1.55 & 1.23 & $-$ & $-$ & 0.49 \\
	     $\delta$~Cep & M97 & 2~449~566.6000 & 0.67 & 0.032 & $ 1.376 \pm 0.620 $ & 3.60 & 3.25 & $-$ & $-$ & 2.51 & 2.27 & 2.22 \\
	     $\delta$~Cep & M97 & 2~449~572.5000 & 0.67 & 0.132 & $ 1.534 \pm 0.767 $ & 3.87 & 3.40 & $-$ & $-$ & 2.56 & 2.27 & 2.21 \\
	     $\delta$~Cep & M97 & 2~449~642.3000 & 0.67 & 0.139 & $ 1.440 \pm 0.567 $ & 3.89 & 3.41 & $-$ & $-$ & 2.56 & 2.27 & 2.21 \\
	     $\delta$~Cep & M97 & 2~449~643.3000 & 0.67 & 0.325 & $ 2.070 \pm 0.431 $ & 4.33 & 3.68 & $-$ & $-$ & 2.63 & 2.28 & 2.21 \\
	     $\delta$~Cep & M97 & 2~449~541.6000 & 0.67 & 0.374 & $ 1.776 \pm 0.504 $ & 4.42 & 3.74 & $-$ & $-$ & 2.66 & 2.28 & 2.22 \\
	     $\delta$~Cep & M97 & 2~449~569.5000 & 0.67 & 0.573 & $ 1.660 \pm 0.504 $ & 4.71 & 3.93 & $-$ & $-$ & 2.76 & 2.36 & 2.28 \\
	     $\delta$~Cep & M97 & 2~449~570.5000 & 0.67 & 0.759 & $ 1.324 \pm 0.841 $ & 4.83 & 4.05 & $-$ & $-$ & 2.87 & 2.49 & 2.40 \\
	     $\delta$~Cep & M97 & 2~449~640.3000 & 0.67 & 0.766 & $ 1.797 \pm 0.462 $ & 4.82 & 4.04 & $-$ & $-$ & 2.87 & 2.49 & 2.40 \\
	     $\delta$~Cep & M97 & 2~449~571.5000 & 0.67 & 0.945 & $ 1.671 \pm 0.431 $ & 3.70 & 3.33 & $-$ & $-$ & 2.58 & 2.34 & 2.30 \\
	     $\delta$~Cep & N02 & 2~450~788.6300 & 0.74 & 0.754 & $ 1.621 \pm 0.063 $ & 4.83 & 4.05 & $-$ & $-$ & 2.87 & 2.48 & 2.40 \\
	     $\delta$~Cep & N02 & 2~450~994.9100 & 0.74 & 0.193 & $ 1.705 \pm 0.094 $ & 4.02 & 3.48 & $-$ & $-$ & 2.58 & 2.27 & 2.21 \\
	     $\delta$~Cep & N02 & 2~450~995.9300 & 0.74 & 0.383 & $ 1.485 \pm 0.115 $ & 4.44 & 3.75 & $-$ & $-$ & 2.66 & 2.29 & 2.22 \\
	     $\delta$~Cep & N02 & 2~450~996.9700 & 0.74 & 0.577 & $ 1.548 \pm 0.220 $ & 4.71 & 3.94 & $-$ & $-$ & 2.76 & 2.37 & 2.29 \\
	     $\delta$~Cep & N02 & 2~450~997.9300 & 0.74 & 0.756 & $ 1.422 \pm 0.115 $ & 4.83 & 4.05 & $-$ & $-$ & 2.87 & 2.48 & 2.40 \\
	     $\delta$~Cep & N02 & 2~450~998.9300 & 0.74 & 0.942 & $ 1.328 \pm 0.125 $ & 3.72 & 3.35 & $-$ & $-$ & 2.59 & 2.35 & 2.30 \\
	     $\delta$~Cep & N02 & 2~451~007.9600 & 0.74 & 0.625 & $ 1.590 \pm 0.105 $ & 4.78 & 4.00 & $-$ & $-$ & 2.81 & 2.40 & 2.32 \\
	     $\delta$~Cep & N02 & 2~451~008.9200 & 0.74 & 0.804 & $ 1.391 \pm 0.084 $ & 4.72 & 3.99 & $-$ & $-$ & 2.86 & 2.49 & 2.41 \\
	     $\delta$~Cep & N02 & 2~451~009.9600 & 0.74 & 0.998 & $ 1.548 \pm 0.073 $ & 3.56 & 3.22 & $-$ & $-$ & 2.52 & 2.29 & 2.24 \\
	     $\delta$~Cep & N02 & 2~451~010.9200 & 0.74 & 0.177 & $ 1.610 \pm 0.073 $ & 3.98 & 3.46 & $-$ & $-$ & 2.57 & 2.27 & 2.21 \\
	     $\delta$~Cep & N02 & 2~451~011.9100 & 0.74 & 0.361 & $ 1.537 \pm 0.073 $ & 4.40 & 3.72 & $-$ & $-$ & 2.65 & 2.28 & 2.21 \\
	     $\delta$~Cep & N02 & 2~451~012.9000 & 0.74 & 0.546 & $ 1.569 \pm 0.073 $ & 4.66 & 3.90 & $-$ & $-$ & 2.73 & 2.34 & 2.27 \\
	     $\delta$~Cep & N02 & 2~451~088.8100 & 0.74 & 0.691 & $ 1.380 \pm 0.125 $ & 4.83 & 4.05 & $-$ & $-$ & 2.85 & 2.45 & 2.36 \\
	     $\delta$~Cep & N02 & 2~451~089.7800 & 0.74 & 0.872 & $ 1.527 \pm 0.073 $ & 4.27 & 3.73 & $-$ & $-$ & 2.76 & 2.44 & 2.38 \\
	     $\delta$~Cep & N02 & 2~451~093.7600 & 0.74 & 0.614 & $ 1.475 \pm 0.021 $ & 4.77 & 3.98 & $-$ & $-$ & 2.80 & 2.39 & 2.31 \\
	     $\delta$~Cep & N02 & 2~451~097.7800 & 0.74 & 0.363 & $ 1.537 \pm 0.052 $ & 4.40 & 3.72 & $-$ & $-$ & 2.65 & 2.28 & 2.21 \\
	     $\delta$~Cep & N02 & 2~451~098.8500 & 0.74 & 0.562 & $ 1.694 \pm 0.063 $ & 4.69 & 3.92 & $-$ & $-$ & 2.75 & 2.36 & 2.28 \\
	     X~Sgr & K04 & 2~452~741.9033 & 2.18 & 0.560 & $ 1.487 \pm 0.058 $ & 4.88 & 4.17 & $-$ & 3.44 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~742.8848 & 2.18 & 0.700 & $ 1.541 \pm 0.067 $ & 4.91 & 4.21 & $-$ & 3.45 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~743.8965 & 2.18 & 0.844 & $ 1.443 \pm 0.065 $ & 4.48 & 3.94 & $-$ & 3.20 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~744.8676 & 2.18 & 0.983 & $ 1.489 \pm 0.059 $ & 4.02 & 3.62 & $-$ & 3.08 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~747.8477 & 2.18 & 0.408 & $ 1.528 \pm 0.217 $ & 4.65 & 4.00 & $-$ & 3.34 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~749.8324 & 2.18 & 0.691 & $ 1.457 \pm 0.104 $ & 4.92 & 4.21 & $-$ & 3.46 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~766.8110 & 2.18 & 0.112 & $ 1.420 \pm 0.078 $ & 4.18 & 3.70 & $-$ & 3.15 & $-$ & $-$ & $-$ \\
	     X~Sgr & K04 & 2~452~768.8768 & 2.18 & 0.406 & $ 1.441 \pm 0.032 $ & 4.65 & 4.00 & $-$ & 3.34 & $-$ & $-$ & $-$ \\
	     $\eta$~Aql & N02 & 2~450~638.8600 & 0.74 & 0.990 & $ 1.959 \pm 0.084 $ & 3.48 & 3.02 & $-$ & 2.51 & 2.23 & 1.95 & 1.90 \\
	     $\eta$~Aql & N02 & 2~450~640.8800 & 0.74 & 0.271 & $ 1.781 \pm 0.094 $ & 3.93 & 3.27 & $-$ & 2.61 & 2.26 & 1.89 & 1.82 \\
	     $\eta$~Aql & N02 & 2~450~641.8600 & 0.74 & 0.408 & $ 1.938 \pm 0.084 $ & 4.24 & 3.47 & $-$ & 2.73 & 2.33 & 1.93 & 1.85 \\
	     $\eta$~Aql & N02 & 2~450~997.8300 & 0.74 & 0.008 & $ 1.697 \pm 0.199 $ & 3.48 & 3.02 & $-$ & 2.51 & 2.22 & 1.94 & 1.89 \\
	     $\eta$~Aql & N02 & 2~450~998.8800 & 0.74 & 0.154 & $ 1.781 \pm 0.063 $ & 3.86 & 3.23 & $-$ & 2.60 & 2.26 & 1.92 & 1.84 \\
	     $\eta$~Aql & N02 & 2~451~007.8800 & 0.74 & 0.408 & $ 1.917 \pm 0.063 $ & 4.24 & 3.47 & $-$ & 2.73 & 2.33 & 1.93 & 1.85 \\
	     $\eta$~Aql & N02 & 2~451~008.9100 & 0.74 & 0.552 & $ 1.603 \pm 0.210 $ & 4.53 & 3.66 & $-$ & 2.89 & 2.43 & 1.99 & 1.90 \\
	     $\eta$~Aql & N02 & 2~451~009.8500 & 0.74 & 0.683 & $ 1.771 \pm 0.084 $ & 4.66 & 3.80 & $-$ & 3.00 & 2.51 & 2.08 & 2.00 \\
	     $\eta$~Aql & N02 & 2~451~010.8400 & 0.74 & 0.821 & $ 1.456 \pm 0.084 $ & 4.25 & 3.55 & $-$ & 2.87 & 2.49 & 2.12 & 2.04 \\
	     $\eta$~Aql & N02 & 2~451~011.8400 & 0.74 & 0.960 & $ 1.509 \pm 0.073 $ & 3.52 & 3.05 & $-$ & 2.54 & 2.25 & 1.97 & 1.92 \\
	     $\eta$~Aql & N02 & 2~451~012.8700 & 0.74 & 0.104 & $ 1.603 \pm 0.105 $ & 3.73 & 3.16 & $-$ & 2.57 & 2.24 & 1.92 & 1.86 \\
	     $\eta$~Aql & L02 & 2~452~065.4200 & 1.64 & 0.764 & $ 1.694 \pm 0.011 $ & 4.53 & 3.73 & $-$ & 2.98 & 2.53 & 2.13 & 2.05 \\
	     $\eta$~Aql & L02 & 2~452~066.4140 & 1.64 & 0.903 & $ 1.694 \pm 0.017 $ & 3.75 & 3.22 & $-$ & 2.65 & 2.34 & 2.03 & 1.97 \\
	     $\eta$~Aql & L02 & 2~452~067.4050 & 1.64 & 0.041 & $ 1.735 \pm 0.041 $ & 3.54 & 3.05 & $-$ & 2.52 & 2.22 & 1.93 & 1.87 \\
	     $\eta$~Aql & L02 & 2~452~075.3830 & 1.64 & 0.153 & $ 1.783 \pm 0.028 $ & 3.86 & 3.23 & $-$ & 2.60 & 2.26 & 1.92 & 1.85 \\
	     $\eta$~Aql & L02 & 2~452~076.3840 & 1.64 & 0.292 & $ 1.843 \pm 0.014 $ & 3.95 & 3.28 & $-$ & 2.61 & 2.26 & 1.89 & 1.82 \\
	     $\eta$~Aql & L02 & 2~452~077.3720 & 1.64 & 0.430 & $ 1.867 \pm 0.022 $ & 4.30 & 3.51 & $-$ & 2.76 & 2.35 & 1.94 & 1.86 \\
	     $\eta$~Aql & L02 & 2~452~089.3500 & 1.64 & 0.099 & $ 1.757 \pm 0.019 $ & 3.71 & 3.15 & $-$ & 2.56 & 2.24 & 1.92 & 1.86 \\
\hline
     $\eta$~Aql & L02 & 2~452~090.3540 & 1.64 & 0.239 & $ 1.842 \pm 0.020 $ & 3.92 & 3.26 & $-$ & 2.61 & 2.26 & 1.90 & 1.83 \\
	     $\eta$~Aql & L02 & 2~452~091.3460 & 1.64 & 0.377 & $ 1.807 \pm 0.023 $ & 4.14 & 3.41 & $-$ & 2.69 & 2.30 & 1.92 & 1.84 \\
	     $\eta$~Aql & L02 & 2~452~095.3600 & 1.64 & 0.936 & $ 1.605 \pm 0.050 $ & 3.59 & 3.10 & $-$ & 2.58 & 2.28 & 1.99 & 1.94 \\
	     $\eta$~Aql & L02 & 2~452~099.3370 & 1.64 & 0.490 & $ 1.844 \pm 0.026 $ & 4.44 & 3.60 & $-$ & 2.83 & 2.40 & 1.97 & 1.88 \\
	     $\eta$~Aql & L02 & 2~452~101.3290 & 1.64 & 0.768 & $ 1.672 \pm 0.038 $ & 4.51 & 3.72 & $-$ & 2.97 & 2.53 & 2.13 & 2.05 \\
	     $\eta$~Aql & L02 & 2~452~103.2930 & 1.64 & 0.042 & $ 1.697 \pm 0.041 $ & 3.54 & 3.05 & $-$ & 2.52 & 2.22 & 1.93 & 1.87 \\
	     $\eta$~Aql & L02 & 2~452~105.3000 & 1.64 & 0.321 & $ 1.842 \pm 0.025 $ & 3.99 & 3.31 & $-$ & 2.63 & 2.27 & 1.90 & 1.82 \\
	     $\eta$~Aql & L02 & 2~452~106.2830 & 1.64 & 0.458 & $ 1.860 \pm 0.016 $ & 4.38 & 3.56 & $-$ & 2.79 & 2.37 & 1.95 & 1.86 \\
	     $\eta$~Aql & L02 & 2~452~107.3020 & 1.64 & 0.600 & $ 1.853 \pm 0.028 $ & 4.59 & 3.71 & $-$ & 2.93 & 2.46 & 2.02 & 1.93 \\
	     $\eta$~Aql & L02 & 2~452~108.3080 & 1.64 & 0.740 & $ 1.744 \pm 0.033 $ & 4.60 & 3.77 & $-$ & 3.00 & 2.54 & 2.13 & 2.04 \\
	     $\eta$~Aql & L02 & 2~452~116.2760 & 1.64 & 0.851 & $ 1.650 \pm 0.024 $ & 4.07 & 3.43 & $-$ & 2.79 & 2.44 & 2.09 & 2.02 \\
	     $\eta$~Aql & K04 & 2~452~524.5643 & 2.18 & 0.741 & $ 1.782 \pm 0.124 $ & 4.60 & 3.77 & $-$ & 3.00 & 2.54 & 2.13 & 2.04 \\
	     $\eta$~Aql & K04 & 2~452~557.5462 & 2.18 & 0.336 & $ 1.916 \pm 0.105 $ & 4.03 & 3.34 & $-$ & 2.64 & 2.28 & 1.90 & 1.83 \\
	     $\eta$~Aql & K04 & 2~452~559.5346 & 2.18 & 0.614 & $ 1.843 \pm 0.045 $ & 4.60 & 3.73 & $-$ & 2.94 & 2.46 & 2.03 & 1.94 \\
	     $\eta$~Aql & K04 & 2~452~564.5321 & 2.18 & 0.310 & $ 1.846 \pm 0.053 $ & 3.97 & 3.30 & $-$ & 2.62 & 2.26 & 1.89 & 1.82 \\
	     $\eta$~Aql & K04 & 2~452~565.5155 & 2.18 & 0.447 & $ 1.910 \pm 0.032 $ & 4.35 & 3.54 & $-$ & 2.78 & 2.36 & 1.95 & 1.86 \\
	     $\eta$~Aql & K04 & 2~452~566.5185 & 2.18 & 0.587 & $ 1.900 \pm 0.034 $ & 4.57 & 3.70 & $-$ & 2.92 & 2.45 & 2.01 & 1.92 \\
	     $\eta$~Aql & K04 & 2~452~567.5232 & 2.18 & 0.727 & $ 1.839 \pm 0.040 $ & 4.63 & 3.79 & $-$ & 3.00 & 2.54 & 2.12 & 2.04 \\
	     $\eta$~Aql & K04 & 2~452~573.5114 & 2.18 & 0.561 & $ 1.923 \pm 0.057 $ & 4.54 & 3.67 & $-$ & 2.90 & 2.43 & 2.00 & 1.91 \\
	     $\eta$~Aql & K04 & 2~452~769.9372 & 2.18 & 0.931 & $ 1.681 \pm 0.031 $ & 3.61 & 3.12 & $-$ & 2.59 & 2.29 & 2.00 & 1.95 \\
	     $\eta$~Aql & K04 & 2~452~770.9222 & 2.18 & 0.068 & $ 1.828 \pm 0.049 $ & 3.61 & 3.09 & $-$ & 2.54 & 2.23 & 1.92 & 1.86 \\
	     $\eta$~Aql & K04 & 2~452~772.8988 & 2.18 & 0.343 & $ 1.919 \pm 0.051 $ & 4.04 & 3.35 & $-$ & 2.65 & 2.28 & 1.90 & 1.83 \\
	     W~Sgr & K04 & 2~452~743.8373 & 2.18 & 0.571 & $ 1.438 \pm 0.103 $ & 5.43 & 4.59 & $-$ & 3.78 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~744.9149 & 2.18 & 0.713 & $ 1.319 \pm 0.094 $ & 5.55 & 4.72 & $-$ & 3.91 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~749.8676 & 2.18 & 0.365 & $ 1.289 \pm 0.147 $ & 5.00 & 4.32 & $-$ & 3.59 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~751.8659 & 2.18 & 0.628 & $ 1.348 \pm 0.179 $ & 5.52 & 4.66 & $-$ & 3.86 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~763.8884 & 2.18 & 0.211 & $ 1.311 \pm 0.035 $ & 4.75 & 4.15 & $-$ & 3.51 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~764.8766 & 2.18 & 0.341 & $ 1.383 \pm 0.030 $ & 4.88 & 4.24 & $-$ & 3.55 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~765.8802 & 2.18 & 0.473 & $ 1.341 \pm 0.033 $ & 5.24 & 4.46 & $-$ & 3.71 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~767.8671 & 2.18 & 0.735 & $ 1.234 \pm 0.083 $ & 5.52 & 4.71 & $-$ & 3.91 & $-$ & $-$ & $-$ \\
	     W~Sgr & K04 & 2~452~769.9137 & 2.18 & 0.005 & $ 1.266 \pm 0.064 $ & 4.35 & 3.92 & $-$ & 3.42 & $-$ & $-$ & $-$ \\
	     $\beta$~Dor & K04 & 2~452~215.7953 & 2.18 & 0.161 & $ 1.884 \pm 0.082 $ & 4.21 & 3.46 & 3.13 & 2.73 & 2.30 & 1.91 & 1.86 \\
	     $\beta$~Dor & K04 & 2~452~216.7852 & 2.18 & 0.261 & $ 1.999 \pm 0.048 $ & 4.40 & 3.58 & 3.22 & 2.80 & 2.34 & 1.92 & 1.86 \\
	     $\beta$~Dor & K04 & 2~452~247.7611 & 2.18 & 0.408 & $ 1.965 \pm 0.060 $ & 4.76 & 3.81 & 3.42 & 2.97 & 2.45 & 1.99 & 1.93 \\
	     $\beta$~Dor & K04 & 2~452~308.6448 & 2.18 & 0.594 & $ 1.886 \pm 0.076 $ & 4.73 & 3.85 & 3.46 & 3.03 & 2.53 & 2.11 & 2.04 \\
	     $\beta$~Dor & K04 & 2~452~567.8272 & 2.18 & 0.927 & $ 1.834 \pm 0.063 $ & 4.02 & 3.39 & 3.09 & 2.73 & 2.35 & 2.01 & 1.95 \\
	     $\beta$~Dor & K04 & 2~452~744.5645 & 2.18 & 0.884 & $ 1.770 \pm 0.072 $ & 4.12 & 3.47 & 3.16 & 2.79 & 2.39 & 2.03 & 1.98 \\
	     $\beta$~Dor & K04 & 2~452~749.5139 & 2.18 & 0.387 & $ 1.965 \pm 0.110 $ & 4.71 & 3.78 & 3.39 & 2.95 & 2.43 & 1.98 & 1.91 \\
	     $\beta$~Dor & K04 & 2~452~750.5111 & 2.18 & 0.488 & $ 1.907 \pm 0.076 $ & 4.83 & 3.91 & 3.48 & 3.04 & 2.52 & 2.05 & 1.98 \\
	     $\beta$~Dor & K04 & 2~452~751.5186 & 2.18 & 0.591 & $ 1.999 \pm 0.172 $ & 4.74 & 3.86 & 3.47 & 3.04 & 2.53 & 2.10 & 2.04 \\
	     $\zeta$~Gem & N02 & 2~451~098.9800 & 0.74 & 0.477 & $ 1.566 \pm 0.221 $ & 5.06 & 4.11 & $-$ & 3.19 & 2.65 & $-$ & 2.13 \\
	     $\zeta$~Gem & N02 & 2~451~229.8300 & 0.74 & 0.368 & $ 1.461 \pm 0.231 $ & 4.95 & 4.02 & $-$ & 3.10 & 2.59 & $-$ & 2.08 \\
	     $\zeta$~Gem & N02 & 2~451~232.7200 & 0.74 & 0.652 & $ 1.619 \pm 0.053 $ & 4.82 & 3.96 & $-$ & 3.18 & 2.67 & $-$ & 2.18 \\
	     $\zeta$~Gem & N02 & 2~451~233.7100 & 0.74 & 0.750 & $ 1.514 \pm 0.063 $ & 4.61 & 3.85 & $-$ & 3.10 & 2.63 & $-$ & 2.18 \\
	     $\zeta$~Gem & L02 & 2~451~605.2260 & 1.64 & 0.349 & $ 1.720 \pm 0.015 $ & 4.91 & 4.00 & $-$ & 3.08 & 2.58 & $-$ & 2.08 \\
	     $\zeta$~Gem & L02 & 2~451~206.2410 & 1.64 & 0.044 & $ 1.719 \pm 0.048 $ & 4.32 & 3.65 & $-$ & 2.92 & 2.48 & $-$ & 2.07 \\
	     $\zeta$~Gem & L02 & 2~451~214.1920 & 1.64 & 0.827 & $ 1.845 \pm 0.062 $ & 4.46 & 3.77 & $-$ & 3.05 & 2.58 & $-$ & 2.16 \\
	     $\zeta$~Gem & L02 & 2~451~615.1800 & 1.64 & 0.330 & $ 1.783 \pm 0.032 $ & 4.88 & 3.97 & $-$ & 3.06 & 2.57 & $-$ & 2.07 \\
	     $\zeta$~Gem & L02 & 2~451~617.1670 & 1.64 & 0.525 & $ 1.629 \pm 0.029 $ & 5.04 & 4.10 & $-$ & 3.21 & 2.67 & $-$ & 2.15 \\
	     $\zeta$~Gem & L02 & 2~451~618.1430 & 1.64 & 0.622 & $ 1.575 \pm 0.008 $ & 4.89 & 4.00 & $-$ & 3.19 & 2.67 & $-$ & 2.18 \\
	     $\zeta$~Gem & L02 & 2~451~619.1680 & 1.64 & 0.723 & $ 1.590 \pm 0.018 $ & 4.67 & 3.88 & $-$ & 3.12 & 2.64 & $-$ & 2.18 \\
	     $\zeta$~Gem & L02 & 2~451~620.1690 & 1.64 & 0.821 & $ 1.627 \pm 0.029 $ & 4.47 & 3.77 & $-$ & 3.06 & 2.58 & $-$ & 2.16 \\
	     $\zeta$~Gem & L02 & 2~451~622.1980 & 1.64 & 0.021 & $ 1.717 \pm 0.047 $ & 4.30 & 3.64 & $-$ & 2.93 & 2.48 & $-$ & 2.07 \\
	     $\zeta$~Gem & L02 & 2~451~643.1610 & 1.64 & 0.086 & $ 1.707 \pm 0.012 $ & 4.37 & 3.68 & $-$ & 2.91 & 2.48 & $-$ & 2.05 \\
	     $\zeta$~Gem & L02 & 2~451~981.1820 & 1.64 & 0.386 & $ 1.730 \pm 0.014 $ & 4.98 & 4.04 & $-$ & 3.12 & 2.60 & $-$ & 2.09 \\
	     $\zeta$~Gem & L02 & 2~451~982.1640 & 1.64 & 0.482 & $ 1.679 \pm 0.021 $ & 5.06 & 4.11 & $-$ & 3.20 & 2.65 & $-$ & 2.13 \\
\hline
	     $\zeta$~Gem & L02 & 2~451~983.2010 & 1.64 & 0.584 & $ 1.631 \pm 0.022 $ & 4.96 & 4.05 & $-$ & 3.21 & 2.67 & $-$ & 2.17 \\
	     $\zeta$~Gem & L02 & 2~451~894.3870 & 1.64 & 0.835 & $ 1.662 \pm 0.020 $ & 4.45 & 3.76 & $-$ & 3.05 & 2.57 & $-$ & 2.16 \\
	     $\zeta$~Gem & L02 & 2~451~895.3690 & 1.64 & 0.932 & $ 1.672 \pm 0.014 $ & 4.32 & 3.66 & $-$ & 2.98 & 2.52 & $-$ & 2.11 \\
	     $\zeta$~Gem & K04 & 2~452~214.8787 & 2.18 & 0.408 & $ 1.715 \pm 0.059 $ & 5.01 & 4.07 & $-$ & 3.14 & 2.62 & $-$ & 2.10 \\
	     $\zeta$~Gem & K04 & 2~452~216.8357 & 2.18 & 0.600 & $ 1.751 \pm 0.088 $ & 4.93 & 4.03 & $-$ & 3.21 & 2.67 & $-$ & 2.18 \\
	     $\zeta$~Gem & K04 & 2~451~527.9722 & 2.12 & 0.739 & $ 1.643 \pm 0.334 $ & 4.63 & 3.86 & $-$ & 3.11 & 2.63 & $-$ & 2.18 \\
	     $\zeta$~Gem & K04 & 2~451~601.8285 & 2.12 & 0.014 & $ 1.748 \pm 0.086 $ & 4.30 & 3.64 & $-$ & 2.93 & 2.48 & $-$ & 2.08 \\
	     $\zeta$~Gem & K04 & 2~451~259.7790 & 2.12 & 0.318 & $ 2.087 \pm 0.291 $ & 4.85 & 3.95 & $-$ & 3.05 & 2.57 & $-$ & 2.06 \\
	     $\zeta$~Gem & K04 & 2~451~262.7400 & 2.12 & 0.610 & $ 1.730 \pm 0.273 $ & 4.91 & 4.02 & $-$ & 3.20 & 2.67 & $-$ & 2.18 \\
	     $\zeta$~Gem & K04 & 2~451~595.8520 & 2.12 & 0.426 & $ 1.423 \pm 0.284 $ & 5.03 & 4.08 & $-$ & 3.15 & 2.63 & $-$ & 2.11 \\
	     $\zeta$~Gem & K04 & 2~451~602.7640 & 2.12 & 0.107 & $ 1.910 \pm 0.216 $ & 4.40 & 3.69 & $-$ & 2.91 & 2.48 & $-$ & 2.05 \\
	     Y~Oph & K04 & 2~452~742.9056 & 2.18 & 0.601 & $ 1.462 \pm 0.120 $ & 4.93 & 4.10 & 3.86 & 3.29 & 2.96 & 2.58 & 2.52 \\
	     Y~Oph & K04 & 2~452~750.8842 & 2.18 & 0.067 & $ 1.414 \pm 0.106 $ & 4.26 & 3.64 & 3.50 & 3.00 & 2.77 & 2.49 & 2.44 \\
	     Y~Oph & K04 & 2~452~772.8308 & 2.18 & 0.349 & $ 1.478 \pm 0.057 $ & 4.75 & 3.95 & 3.72 & 3.16 & 2.86 & 2.50 & 2.43 \\
	     Y~Oph & K04 & 2~452~786.8739 & 2.18 & 0.168 & $ 1.436 \pm 0.046 $ & 4.45 & 3.76 & 3.59 & 3.07 & 2.79 & 2.48 & 2.42 \\
	     $\ell$~Car & K04 & 2~452~453.4978 & 2.18 & 0.587 & $ 3.035 \pm 0.109 $ & 4.73 & 3.46 & 2.89 & 2.41 & 1.72 & 1.16 & 1.06 \\
	     $\ell$~Car & K04 & 2~452~739.5644 & 2.18 & 0.634 & $ 2.859 \pm 0.084 $ & 4.73 & 3.48 & 2.91 & 2.42 & 1.74 & 1.19 & 1.09 \\
	     $\ell$~Car & K04 & 2~452~740.5691 & 2.18 & 0.662 & $ 2.954 \pm 0.046 $ & 4.72 & 3.47 & 2.92 & 2.42 & 1.75 & 1.22 & 1.12 \\
	     $\ell$~Car & K04 & 2~452~741.7171 & 2.18 & 0.694 & $ 2.969 \pm 0.038 $ & 4.71 & 3.46 & 2.93 & 2.42 & 1.77 & 1.24 & 1.15 \\
	     $\ell$~Car & K04 & 2~452~742.7009 & 2.18 & 0.722 & $ 2.874 \pm 0.054 $ & 4.69 & 3.45 & 2.94 & 2.43 & 1.80 & 1.27 & 1.18 \\
	     $\ell$~Car & K04 & 2~452~743.6985 & 2.18 & 0.750 & $ 2.737 \pm 0.073 $ & 4.66 & 3.44 & 2.94 & 2.44 & 1.81 & 1.30 & 1.21 \\
	     $\ell$~Car & K04 & 2~452~744.6336 & 2.18 & 0.776 & $ 2.769 \pm 0.033 $ & 4.62 & 3.43 & 2.92 & 2.43 & 1.82 & 1.31 & 1.22 \\
	     $\ell$~Car & K04 & 2~452~745.6285 & 2.18 & 0.804 & $ 2.652 \pm 0.095 $ & 4.53 & 3.39 & 2.88 & 2.41 & 1.81 & 1.32 & 1.23 \\
	     $\ell$~Car & K04 & 2~452~746.6198 & 2.18 & 0.832 & $ 2.749 \pm 0.045 $ & 4.40 & 3.33 & 2.81 & 2.37 & 1.79 & 1.31 & 1.23 \\
	     $\ell$~Car & K04 & 2~452~747.5988 & 2.18 & 0.860 & $ 2.674 \pm 0.125 $ & 4.22 & 3.23 & 2.73 & 2.30 & 1.74 & 1.29 & 1.21 \\
	     $\ell$~Car & K04 & 2~452~749.5763 & 2.18 & 0.915 & $ 2.620 \pm 0.076 $ & 3.82 & 2.96 & 2.53 & 2.13 & 1.63 & 1.23 & 1.15 \\
	     $\ell$~Car & K04 & 2~452~751.5785 & 2.18 & 0.972 & $ 2.726 \pm 0.032 $ & 3.62 & 2.79 & 2.40 & 2.01 & 1.54 & 1.15 & 1.08 \\
	     $\ell$~Car & K04 & 2~452~755.6166 & 2.18 & 0.085 & $ 2.942 \pm 0.110 $ & 3.78 & 2.85 & 2.44 & 2.01 & 1.51 & 1.06 & 0.98 \\
	     $\ell$~Car & K04 & 2~452~763.5551 & 2.18 & 0.309 & $ 3.157 \pm 0.032 $ & 4.31 & 3.13 & 2.63 & 2.15 & 1.53 & 1.01 & 0.92 \\
	     $\ell$~Car & K04 & 2~452~765.5545 & 2.18 & 0.365 & $ 3.175 \pm 0.033 $ & 4.43 & 3.19 & 2.69 & 2.21 & 1.56 & 1.02 & 0.93 \\
	     $\ell$~Car & K04 & 2~452~766.5497 & 2.18 & 0.393 & $ 3.173 \pm 0.033 $ & 4.48 & 3.25 & 2.72 & 2.23 & 1.57 & 1.04 & 0.95 \\
	     $\ell$~Car & K04 & 2~452~768.5663 & 2.18 & 0.450 & $ 3.155 \pm 0.036 $ & 4.60 & 3.35 & 2.78 & 2.29 & 1.62 & 1.07 & 0.98 \\
	     $\ell$~Car & K04 & 2~452~769.5746 & 2.18 & 0.478 & $ 3.155 \pm 0.021 $ & 4.65 & 3.39 & 2.81 & 2.32 & 1.64 & 1.08 & 1.00 \\
	     $\ell$~Car & K04 & 2~452~770.5353 & 2.18 & 0.505 & $ 3.124 \pm 0.021 $ & 4.69 & 3.41 & 2.84 & 2.35 & 1.66 & 1.10 & 1.01 \\
	     $\ell$~Car & K04 & 2~452~771.5281 & 2.18 & 0.533 & $ 3.100 \pm 0.019 $ & 4.72 & 3.43 & 2.86 & 2.38 & 1.69 & 1.12 & 1.03 \\
	     $\ell$~Car & K04 & 2~452~786.6200 & 2.18 & 0.957 & $ 2.700 \pm 0.064 $ & 3.64 & 2.82 & 2.42 & 2.03 & 1.56 & 1.17 & 1.09 \\
\hline 
\end{tabular}
\end{table}